Author’s response to reviews

Title: Including pork in the Mediterranean diet for an Australian population: Protocol for a randomised controlled trial assessing cardiovascular risk and cognitive function

Authors:
Alexandra Wade (alexandra.wade@mymail.unisa.edu.au)
Courtney Davis (courtney.davis@unisa.edu.au)
Kathryn Dyer (kate.dyer@unisa.edu.au)
Jonathan Hodgson (jonathan.hodgson@uwa.edu.au)
Richard Woodman (richard.woodman@flinders.edu.au)
Hannah Keage (Hannah.Keage@unisa.edu.au)
Karen Murphy (karen.murphy@unisa.edu.au)

Version: 1 Date: 17 Nov 2017

Author’s response to reviews:

We would like to thank each of the reviewers for their consideration of our manuscript and for their constructive feedback.

Please find attached our responses to each of the reviewers’ comments and details of any changes made where applicable. We have attempted to incorporate all feedback into the revised protocol paper where appropriate.

Thank you for considering our manuscript for publication.